Author's response to reviews

Title: A forced titration study of the antioxidant and immunomodulatory effects of Ambrotose AO supplement

Authors:

Stephen P Myers (smyers@scu.edu.au)
Lesley Stevenson (LStevenson@hortresearch.co.nz)
Phillip A Cheras (philcheras@yahoo.com.au)
Joan O'Conner (joan.oconnor@scu.edu.au)
Lyndon Brooks (lyndon.brooks@scu.edu.au)
Margaret Rolfe (margaret rolfe@scu.edu.au)
Paul Conellan (paul.conellan@scu.edu.au)
Carol Morris (carol.morris@scu.edu.au)

Version: 5 Date: 22 April 2010

Author's response to reviews:

Dear Editors,

I have made one correction to author affiliations due to a change in research entities at Southern Cross University – in affiliation 2 now reads:

NatMed-Research Unit, Research Cluster for Health and Wellbeing, Southern Cross University, Lismore, Australia

I have made three minor corrections to the final manuscript.

1. Methods Section – Participants (Paragraph 3)
   The word ‘not’ was included in the following sentence - “Baseline bloods were also reviewed to ensure subjects with undiagnosed abnormalities were not included”

2. Results Section – Safety (Sentence 1)
   ‘Subjects’ has been changed to ‘participants’

3. Results Section – Safety (Sentence 2)
   ‘Subjects’ has been changed to ‘participants’

With these changes the manuscript is ready for publication.

All the best.

Stephen

Professor Stephen P Myers PhD BMed ND

Director

NatMed-Research

The Natural and Complementary Medicine Research Unit